Author's response to reviews

Title: The UK Burden of Injury Study - a protocol.

Authors:

Ronan A Lyons (r.a.lyons@swansea.ac.uk)
Elizabeth M Towner (Elizabeth.Towner@uwe.ac.uk)
Denise Kendrick (Denise.Kendrick@nottingham.ac.uk)
Nicola Christie (N.Christie@surrey.ac.uk)
Sinead Brophy (s.brophy@swansea.ac.uk)
Ceri J Phillips (c.j.phillips@swansea.ac.uk)
Carol Coupland (Carol.Coupland@nottingham.ac.uk)
Rebecca Carter (Carterbecky@hotmail.com)
Lindsay Groom (Lindsay.Groom@nottingham.ac.uk)
Judith Sleney (j.sleney@surrey.ac.uk)
Phillip Adrian Evans (Phillip.Evans@swansea-tr.wales.nhs.uk)
Ian Palister (I.Palister@swansea.ac.uk)
Frank Coffey (Frank.Coffey@nottingham.ac.uk)

Version: 3 Date: 22 October 2007

Author's response to reviews:

Thank you very much for publishing this protocol. We have made all the changes requested and would like to say thank you for accepting the article.

Changes made:

Title and abstract - These are identical in the manuscript and in the submission system.

Author list - These are written exactly as in the submission system.

Country - the UK has now been added to author 1.

Conclusions has been changed to discussion

Competing interests section has been moved

Author's contributions has been changed and moved

The references has be ammended

The table is has not been divided into parts a b and c and is refered to as table 1 in the text.